

Response to Reviewer and Editor Comments

Dear Editor,

We thank you for the assessment of our manuscript entitled “BioInteract: Interpretable Drug–Target Interaction Prediction via Residue-Level Cross-Attention with Biological Prior Knowledge”. We have carefully addressed the data availability mismatch noted in the decision letter. The revised manuscript and the submission-system entry now use the same Data Availability statement.

Point-by-point response

Comment 1. Data availability mismatch: We observed a discrepancy between the data availability statements provided in the system and in the manuscript. Kindly ensure that an identical statement is used in both locations.

Response: We thank the editor for identifying this mismatch. We have revised the manuscript section heading and statement to ensure that the Data Availability statement in the manuscript is identical to the text provided in the submission system. The revised statement is:

All data used in this study are publicly available. The Davis kinase inhibitor dataset used for model training and evaluation is available from the original publication. The BioInteract source code, pretrained model weights, and figure-generation scripts are available at <https://github.com/Liubo-520/bio-intelligence>. An interactive web server for drug–target prediction and attention-map visualisation is available at <https://huggingface.co/spaces/AI4deeperScience/BioInteract>.

The corresponding changes are highlighted in red in the marked manuscript.

Sincerely,

The authors